Author's response to reviews

Title: Electroacupuncture activates corticotrophin-releasing hormone-containing neurons in the paraventricular nucleus of the hypothalamus to alleviate edema in a rat model of inflammation

Authors:

Aihui Li (ali002@umaryland.edu)
Lixing Lao (llao@compmed.umm.edu)
Yi Wang (wangyi01101971@yahoo.com.cn)
Jiajia Xin (Xinjiajia@hotmail.com)
Ke Ren (kren@umaryland.edu)
Brian M. Berman (bberman@compmed.umm.edu)
Ming Tan (MTan@umm.edu)
Ruixin Zhang (rzhan001@umaryland.edu)

Version: 4 Date: 1 May 2008

Author's response to reviews:

Dear editor:

We formatted the MS according to the comments:

1. Separate the e-mail and the affiliations;
2. The e-mail has been preceded by initials;
3. All text is in black;
4. The key words have been deleted;
5. "Legends" have been changed to "Figure legends"
6. "Fig." has been changed to "Figure"

Sincerely,
Ruixin